Author's response to reviews

Title: Relation between dietary cadmium intake and biomarkers of cadmium exposure in premenopausal women accounting for body iron stores

Authors:

Bettina Julin (bettina.julin@ki.se)
Marie Vahter (marie.vahter@ki.se)
Billy Amzal (billy.amzal@la-ser.com)
Alicja Wolk (alicja.wolk@ki.se)
Marika Berglund (marika.berglund@ki.se)
Agneta Åkesson (agneta.akesson@ki.se)

Version: 6 Date: 12 December 2011

Author’s response to reviews:

The manuscript has now been read very carefully and we discovered two small typographical changes we would like to make.